Author's response to reviews

Title: Effect of Early and Current Helicobacter pylori infection on the risk of anaemia in 6.5-year-old Ethiopian children

Authors:

Bineyam Taye (bineymt@gmail.com)
Fikre Enquselassie (fikreens@yahoo.com)
Aster Tsegaye (tsegayaster@yahoo.com)
Alemayehu Amberbir (alemayehu.mberbir@lshtm.ac.uk)
Girmay Medhin (g.tmedhine@yahoo.com)
Andrew Fogarty (Andrew.Fogerty@nottingham.ac.uk)
Karen Robinson (karen.robinson@nottingham.ac.uk)
Gail Davey (g.davey@bsms.ac.uk)

Version: 2
Date: 6 March 2015

Author's response to reviews:

Dear Editors,

The issues raise on page number is addressed in revised manuscript

With warm regards,

Bineyam